Author's response to reviews

Title: Gene polymorphisms in APOE, NOS3, and LIPC genes may be risk factors for cardiac adverse events after primary CABG

Authors:

Sandra Eifert (seifert@med.uni-muenchen.de)
Astrid Rasch (Astrid.Rasch@med.uni-muenchen.de)
Andres Beiras-Fernandez (abeiras@med.uni-muenchen.de)
Georg Nollert (gnollert@med.uni-muenchen.de)
Bruno Reichart (breichart@med.uni-muenchen.de)
Peter Lohse (plohse@med.uni-muenchen.de)

Version: 4 Date: 11 August 2009

Author's response to reviews:

Dear Editors,

we have, as requested, changed the following points:

- Additional files list has been included in the manuscript.
- The references have been formatted, including the number 16
- Minor revisions have also been done, including the removing of words from the title page, editing of figure 1, and editing of the tables.

We hope you find the manuscript suitable for publication.

Best Regards

Andres Beiras-Fernandez MD, PhD